CASE REPORT Open Access

# Treatment of eating disorders in adolescents during the COVID-19 pandemic: a case series



Serur Yaffa<sup>1,2,3</sup>, Enoch-Levy Adi<sup>1,2,3</sup>, Pessach Itai<sup>3,4</sup>, Joffe-Milstein Marit<sup>1,2,3</sup>, Gothelf Doron<sup>2,3,4,5</sup> and Stein Daniel<sup>1,2,3,4\*</sup>

**Abstract: Background:** Eating disorders (EDs) are among the most difficult psychiatric disorders to treat in normal conditions. They are likely even more difficult to manage in at-risk conditions such as the COVID-19 pandemic. Currently there is limited evidence about the particular needs and recommended treatment of adolescents with EDs during the COVID-19 outbreak, in particular regarding the use of telemedicine and the involvement of the family in long distance-treatment.

**Aims:** We sought to discuss the advantages and problems associated with the use of multi-professional long-distance telemedicine treatment in the management of adolescents with EDs and their families during the COVID-19 outbreak.

**Methods:** We gathered data about the treatment of adolescents with EDs in our pediatric ED-treatment center in Israel during the COVID-19 outbreak in the first 10 months of 2020, and compared it to the respective period in the past five years (2015–2019). Second, we described the management of four young females with anorexia nervosa (AN), treated in the ambulatory, daycare and inpatient facilities of our center during the COVID-19 pandemic.

**Findings:** Slightly less patients were treated in our center during the COVID-19 pandemic than in the respective period in the past five years. These patients received at that time considerably more treatment sessions from all treatment providers (psychiatrists, clinical nutritionists and psychotherapists). This was related, in part, to the extensive use of telemedicine during that period (more than as third of all sessions were carried out with telemedicine in comparison to no use of long-distance treatment in the previous years). The condition of the four adolescents with AN was compromised at the start of the COVID-19 quarantine. The use of multi-disciplinary long-distance telemedicine treatment resulted in an improvement in the condition in three of the four adolescents, living in well-organized families, with the motivation and ability to adjust to the new conditions, but not in one adolescent whose family experienced more problems. These families might require the use of face-to-face interventions even during pandemic conditions.

**Conclusion:** The choice of the mode of treatment for adolescents with EDs during pandemic times (telemedicine vs. face-to-face) should consider the functioning of the family.

**Keywords:** COVID-19, Eating disorders, Pandemic, Telemedicine

Full list of author information is available at the end of the article



© The Author(s). 2021 **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

<sup>\*</sup> Correspondence: prof.daniel.stein@gmail.com

<sup>&</sup>lt;sup>1</sup>Pediatric Psychosomatic Department, Sheba Medical Center, 5265601 Tel Hashomer, Israel

<sup>&</sup>lt;sup>2</sup>Child and Adolescent Psychiatric Division, Sheba Medical Center, 5265601 Tel Hashomer. Israel

# Lay summary

Anorexia nervosa is one of the most complicated psychiatric disorders to treat in normal conditions. It is likely even more difficult to manage in at-risk conditions such as the recent COVID-19 (Corona) outbreak. In this article we describe the management of four young female adolescents with anorexia nervosa during the COVID-19 quarantine in a center treating children and adolescents with eating disorders in Israel. We used long-distance treatment (telemedicine) with a team of dietitians, psychotherapists and psychiatrists to help the parents and patients with eating at home, and with other problems arising from the COVID-19 outbreak and the associated changes in living conditions. The findings show a deterioration in the condition of the four adolescents at the start of the COVID-19 quarantine. They found it difficult to adjust themselves to the greater involvement of their families with their eating, and to the reduction in physical activity because of the required confinement. The use of multi-professional long-distance treatment showed an improvement in the condition of three of the four adolescents, living in well-organized families, with the motivation and ability to adjust to the new conditions, but not for one girl, whose family experienced more problems.

# **Background**

In March 2020, the World Health Organization (WHO) declared the Corona Virus (COVID-19) outbreak in 2020 as a pandemic, affecting economy, healthcare, and well-being. Due to the need for social distancing, inperson visits for clinical care were reduced worldwide, and treatment providers had to learn new skills within a brief period and adapt clinical care to the new conditions [1, 2].

People with psychiatric disorders are affected significantly more by external stressors such as the COVID-19 outbreak, likely due to them using more dysfunctional emotion regulation and coping strategies than healthy individuals [3]. Adults [4, 5] and adolescents [6] with psychiatric problems have been found to show acute emotional reactions to the COVID-19 outbreak, with its consequent quarantine and social isolation, including depressive, anxiety, and trauma-related symptoms [4–6].

# EDs in the time of COVID-19

Anorexia nervosa (AN) is one of the most complicated psychiatric disorders to treat in normal conditions [7]. It is likely even more difficult to manage in at-risk conditions such as the COVID-19 pandemic. This is because the management of these patients usually requires regular clinical follow-ups. However, during the current COVID-19 pandemic, face-to-face clinical assessments carry a particularly high risk of infection and

transmission in this vulnerable cohort, already compromised by low weight and reduced immunity [8, 9].

Recent surveys of adult patients with AN and bulimia nervosa (BN) [4-7, 10-15] during the COVID-19 pandemic found an increase in eating disorder (ED)-related symptoms, heightened overall anxiety, and reduced quality of life. Patients reported of an increase of 39-87% in the frequency of food restriction, binge-eating episodes, and purging behaviors. They also reported fears of loss of routine and control regarding their meal plan, related in part to reduced access to ED treatment services. The provision of adequate care in patients with EDs during periods of acute pandemic and confinement requires both psychological and medical interventions in people who might be medically compromised, resistant to treatment, and suicidal [9, 15-17]. These interventions have to be relatively easy to implement, and not necessarily carried out with face-to-face meetings. They require adaptations for dealing with issues such as home weighing and management of binge/purge behaviors, likely involving multiple interventions from different treatment providers with different family members to adequately supervise eating.

In contrast to the findings in adult patients, there is only limited information about the treatment of adolescent patients with EDs during the COVID-19 outbreak [9, 18, 19]. A recent study [9] in seven adolescent patients with AN suggested a treatment program tailored specifically for the COVID-19 period. The program included long-distance homebased sessions instead of clinic/hospital attendances, biweekly weighing, recording of food intake to allow for estimated calculations of daily calories, estimation of energy expenditure, and collection of vital signs when possible. The clinicians in this study used a structured clinical interview with the parents to make accurate clinical assessments remotely. While there was no description of the efficacy of this program, the authors emphasized its potential utility in maintaining ongoing delivery of treatment and parental support, balancing the protective effect of few or no clinical attendances against the risk of excessive carer burden and unnoticed patient deterioration. In another study [19] carried out during the COVID-19 outbreak, almost half of the children and adolescents with EDs studied, experienced reactivation of their ED symptoms despite treatment, and symptoms of self-harm and increased suicide risk occurred in patients with severe illness (25% of the sample). In that study, telephone and videoconferencing visits were held in both the day hospital and outpatient programs; under this program, certain face-to-face visits continued, while combined treatment was instituted according to clinical indication.

# Telemedicine treatment in the time of COVID-19

The findings of previous studies about the treatment of patients with EDs during the COVID-19 period [9, 19] highlight the importance of adapting to the use of longdistance telemedicine. Telemedicine refers to the provision of remote clinical services, via real-time communication, between patients/families and healthcare providers, using electronic audio and visual means. Telemedicine services may expand access by reducing barriers such as travel time, competing responsibilities, or absence from work, and provide advantages for treatment providers and institutions, including schedule flexibility, increased productivity, and less clinic overhead [2]. With the ongoing COVID-19 pandemic, telemedicine might be also useful in decreasing emergency room visits and safeguarding healthcare resources, potentially reducing the spread of the virus [20].

Moreover, no difference was found in the effectiveness and treatment adherence of face-to-face vs. internetbased interventions in people with mental-health problems [21-24]. Specifically, videoconferencing was as effective as in-person treatment for reducing the severity of post-traumatic, anxiety and depressive symptoms [25–27]. Similarly, cognitive behavioral treatment (CBT) by videoconferencing (telehealth) had good clinical efficacy for ED treatment [28], including in adolescents [29]. Moreover, telehealth was found appropriate also during the COVID-19 pandemic in reducing the mental health burden of patients and families [30-33]. For example, studies comparing telemedicine to face-to-face interventions during the COVID-19 outbreak found positive experiences with both options in caregivers of adult patients with EDs [34]. Last, psychologists who were more likely to use teletherapy prior to COVID-19, being more comfortable with its use, showed more positive attitudes to telepsychology, and in turn, greater use of it, during the pandemic [35].

# The aims of the present study

Given the scarcity of data regarding how to support adolescents with EDs during the COVID-19 pandemic, and the need to adapt treatment to the specific conditions of the resulting mandatory confinement, we sought to analyze our data about the management of adolescent patients with EDs in Israel during the COVID-19 period. Second, we sought to describe four female adolescents with AN treated in the different settings of our pediatric ED-center, emphasizing the obstacles and challenges of the treatment of EDs under these conditions. The four cases described here illustrated the emotional condition of adolescents with AN during the COVID-19 outbreak, the way in which they and their families adapted to the quarantine, and the pros and cons of telemedicine use.

#### **Methods**

# Treatment provision in the present study

The pediatric ED treatment center in the Safra Children's Hospital at the Sheba Medical Center, Tel Hashomer, Israel, includes inpatient, daycare, and ambulatory services for children and adolescents between the ages of 6–18 with diverse types of EDs. Treatment is provided by a multidisciplinary team at varying levels of intensity, depending on the severity of the ED and comorbid disorders, and the overall functioning of the patient and family. The center is a "wrap-around" service, i.e., patients may move from one facility to another, depending on their condition.

During inpatient treatment, patients receive treatment interventions tailored for the treatment of the ED, comorbid disorders, and different psychosocial difficulties. Upon discharge, patients are offered a multidisciplinary daycare program in the afternoon hours, to allow for their reintegration into the school system. When considered stabilized, patients are referred to our ambulatory service. Most of the patients in our ambulatory facility are sent from different community services or are self-referred.

The integrative treatment protocol in our service corresponds with other structured programs for adolescents with AN and BN [36, 37]. The protocol includes the following: a behaviorally oriented nutritional rehabilitation program; either individual psychodynamic-oriented psychotherapy or individual CBT, depending on the specific illness and the aims of treatment; individual expressive movement therapy; family therapy (either family-based treatment (FBT [38]) or systemic family therapies [39]) or parental consultation; and different group interventions: psychodynamic group therapy, CBT group sessions ("classical" ED-related CBT [40] and cognitivemotivational treatment based on the Maudsley Model of Anorexia Nervosa Treatment for Adults (MANTRA) protocol [41], and group expressive movement therapy. The inclusion of psychodynamic psychotherapy in the treatment regimen is designed to address intrapsychic and interpersonal developmental needs of adolescents often burdened with long-standing illness, in addition to the specific EDrelated therapies administered [42]. It is of note that other programs in patients with AN have used psychodynamic psychotherapy as their main treatment, showing favorable results [43]. While inpatient treatment includes all these therapeutic modalities, daycare and ambulatory treatment includes only some of these interventions, as required, but nutritional consultation, individual psychotherapy, family therapy/parental consultation, and some group interventions are usually maintained.

# Data analysis

For this study, we have retrospectively reviewed data about the treatment in our ambulatory service during the COVID-19 period. This data appears in a specific computerized database of the Sheba Medical Center called CHESS. This database includes data about the number of patients, sessions, and treatment providers treating the patients (in our case psychiatrists, clinical nutritionists and psychotherapists) in a certain period. It does not include any other data. We further describe the management of four female adolescents with AN treated in our ambulatory, daycare and inpatient services.

# **Findings**

The first cases of COVID-19 in Israel were reported in February 2020. Starting on March 15th, 2020, and continuing for two months, people were not allowed to be treated frontally unless for emergency care; this required the transfer of most of our services to telemedicine to provide patients and families with adequate support and continuity of clinical care. In the ambulatory service, psychiatric, nutritional and psychological treatment were aimed to be continued as before with specific required adaptations, but carried out via telemedicine. Individual, family therapy/parental counseling and group therapies were, thus, offered online. Nutritionists asked the patients to weigh themselves weekly with the supervision of their parents, if this was possible and agreed upon by both parties. Weighing took place during the morning hours at the patients' home or at the nearest community medical service. Patients were also asked to carry out regular food monitoring. Face-to-face visits occurred only in specific conditions.

All daycare patients, and those inpatients considered physically and emotionally stable, were discharged to their homes. There, they received a long-distance treatment protocol similar to that of the ambulatory services, but were asked to measure weight and vital signs once a week at the hospital (similar to the recommendations of other studies, e.g., Graell et al. [19]). Patients continuing with inpatient treatment were not able to leave the hospital during the two-month COVID-19 lockdown, or to receive visits from their families and peers. Individual sessions with the multidisciplinary treatment providers still continued on a face-to-face basis, but all family treatment/parental consultation sessions were transformed to telemedicine (with the patients and therapists in the department, and the parents and other family members at home).

Telemedicine services enabled maintenance of continuity of care until the end of the quarantine, when face-to-face services could be resumed. Thereafter, some of the ambulatory and daycare treatments continued with telemedicine because of the confinement of

patients, families, or treatment providers, or because of the refusal of patients/families to be managed frontally in a hospital treating patients with COVID-19. Other patients and families resumed face-to-face interventions.

A second quarantine occurred between the 13th of September and the 11th of October 2020. Ambulatory, daycare and stabilized inpatients were again offered home-based long-distance telemedicine treatment, but in contrast to the first lockdown, the decision of keeping the treatment face-to-face vs. via telemedicine could be agreed upon by treatment providers and/patients/families. It should be noted that although possible, we did not treat our patients with telemedicine prior to the COVID-19 outbreak.

The findings emerging from the treatment in our ambulatory service during the COVID-19 pandemic between 01/01/2020–31/10/2020 have been unexpected. Table 1 summarizes the findings for the 2.5 months before the COVID-19 outbreak (01/01/2020–15/03/2020), the 2 months of the first COVID-19-related lockdown (15/03/2020–15/05/2020), the 4 months after the release of the 1st mandatory quarantine (15/05/2020–13/09/2020), the four weeks of the second mandatory quarantine (13/09/2020–11/10/2020), and the next three weeks until 31/10.2020. We have compared this data to the respective period in 2015–2019 (see Table 1).

The findings show that for the period of 01/01-30/10, ambulatory patients and their families have received more multidisciplinary sessions during the time of the COVID-19 (5926 sessions), compared to the respective period between 2015 and 2019 (4001 sessions) from all team members - psychiatrists, clinical nutritionists, and psychotherapists. The number of patients treated in our ambulatory service during these ten months was slightly lower than their mean yearly number in the respective previous period (242 vs. 257); although fewer new cases were accepted at that time, their number was still considerably high. The increase in the number of sessions was accounted for, at least in part, to the possibility of carrying-out multi-professional telemedicine meetings, comprising of 37% of all sessions during the first 10 months of 2020, vs. no use during the respective period between 2015 and 2019 (2192 of 5926 sessions, see Table 1).

# Case reports

Due to the nature of this case-report study, ethics approval was not required by the institution. Demographic and clinical details of the participants were changed to prevent identification. Verbal consent was obtained for publication from patients and parents.

# Ambulatory treatment: case 1

M. is a 13-year old 7th grade female student, the youngest of three children of divorced parents. There is no

Yaffa et al. Journal of Eating Disorders (2021) 9:17 Page 5 of 11

**Table 1** Data of treatment in our ambulatory service during the COVID-19 pandemic and the mean respective period between 2015 and 2019

|  | 2020 (01/01–31/10) | Mean yearly<br>01/01–31/10; from 2015 to 2019 |
|--|--------------------|---|
| Sessions   | 5926               | 4001  |
| Patients   | 242                | 257   |
| New patients   | 127                | 166   |
| Meeting with a psychiatrist  | 662                | 433   |
| Meeting with a clinical nutritionist   | 1722               | 750   |
| Meeting with a mental health professional (psychologist, art/movement therapist) | 3318               | 1690  |
| Telemedicine Sessions (with any team member)                                     | 2192               |   |

evidence of psychiatric history in the family. M. is living with her mother, but visits her father regularly. She is diagnosed with attention deficit/hyperactivity disorder, learning disorders, and generalized anxiety disorder.

M. was referred to our ambulatory service following a short hospitalization in a pediatric department because of severe bradycardia (HR at night ≤44 BPM). She was diagnosed with restricting type AN and a major depressive episode with occasional suicidal thoughts and severe self-injurious behaviors (SIBs). At admission to our service, her body mass index (BMI) was 16.9 kg/m<sup>2</sup>. M. started with fluoxetine up to 30 mg/day and risperidone up to 1.4 mg/day. The treatment was via telemedicine from the start, and included weekly meetings with a clinical nutritionist, psychiatrist and psychologist. She was weighed at home in the presence of her mother. M. was afraid to gain weight and reacted with impulsive SIBs, requiring the presence and support of the nutritionist and psychiatrist via telemedicine during the process of weighing. Directly after weighing, M. had a telemedicine session with her psychologist to help her cope with her anxiety. Additionally, because of the COVID-19 quarantine, her older siblings returned home. Her sister was constantly exercising, and both siblings were supervising and criticizing M's eating, making her angry and less cooperative. Telemedicine sessions enabled flexible treatment settings, which met with M.'s changing needs and abilities and enabled the inclusion of the family to assist her. The siblings were invited to participate in the telemedicine sessions with M. and the psychologist, thus reducing their criticism and increasing their cooperation with the treatment. Sessions with the mother empowered her supervision over her eating.

M. reached the minimal range of her target weight within two months, followed by the return of her menses. She gradually returned to school and meet with her friends. She showed good control over her eating, and her psychotherapy concentrated mainly on the problems she had with her peers. Upon the request of M and her mother, individual psychotherapy has been resumed on a face-to-face basis, whereas the consultations with the

mother (with a different psychologist), family sessions with different family members (carried out by the two psychologists), nutritional consultation (including weighing), and psychiatric consultation (she is currently treated with duloxetine 30 mg/day and risperdone 1 mg/day) are carried out with telemedicine.

# Ambulatory treatment: case 2

N. is a 14 years-old, 9th grade female student, the third of four children, living with her parents and sisters. There is no known family history of psychiatric disorders. Her parents report normal development.

She was diagnosed with AN-purging type with additional rage outbursts, including shouting and the throwing of objects. N. was referred to our ambulatory service following a hospitalization in a pediatric department because of low weight (BMI =  $15.4\,\mathrm{kg/m^2}$ ) and lack of cooperation with her previous ambulatory treatment. During this hospitalization, she was treated with risperidone 0.5 mg/day, with a decrease in the frequency and intensity of the outbursts.

When school was discontinued because of the COVID-19 outbreak, all the family was confined at home. N.'s severe AN-related eating behaviors were brought to the foreground. She ate very slowly, spat out her food and was frequently in the toilet following her meals. The sessions with her psychiatrist, nutritionist and psychologist were transferred to telemedicine. She met with the psychiatrist every two weeks, and had a once-weekly session with her psychologist, and onceweekly session with the nutritionist. The mother was present with the girl in several nutritional counseling sessions, but not in others. In the sessions where only N. was present, she reported of feeling better, denying all her ED-related symptoms, and constantly demanded the stopping of her medications. When the nutritionist heard about the worsening of her symptoms in the sessions where the mother was also present, she recommended a complete supervision of N.s' meals, including supervision on the preparation, eating and avoiding her entering the restroom after meals. As the father

continued working outside the home, and the mother was confined to bed because of severe back pain, it was decided by the team and parents that N's sisters would cook and supervise her meals. The mother felt helpless in this position and asked for more support from the treatment team. As a result, in addition to the regular telemedicine treatment plan, we included a once-weekly separate guidance to the mother.

Our plan was that N. would benefit from the support of all her family at home during the COVID-19 quarantine to rehabilitate her eating. In this case, however, the opposite occurred. The family intervention brought to continuous tensions between N. and her mother and sisters, and N's rage outbursts worsened again. She resented her sisters for telling her what to do, and the mother was feeling helpless despite her separate supervision. The long-distance telemedicine family consultations and individual psychotherapy sessions had no effect on N.'s outbursts and her cooperation with her meal plan.

After gaining one kilogram (BMI = 17.6 kgs/m<sup>2</sup>), N. refused any supervision on her eating, stopped cooperating with the team, and started to lose weight. The family felt that they could not handle N.'s supervision anymore. After the first quarantine ended, N. was admitted to inpatient treatment; currently, she is still hospitalized.

# Daycare treatment: case 3

Y. is a 15 years-old 10th grade female student, diagnosed with AN-restricting type, the second of four children, living with her parents and siblings. There is no known family history of psychiatric disorders. There have been no reported problems during her development.

Y. was referred to our half-way out daycare service following seven months of inpatient treatment, in which she was cooperative and was able to reach her recommended weight. The transition from inpatient treatment to daycare was very hard for Y. She struggled with eating independently, leading to gradual but constant weight loss and irritability. The multidisciplinary treatment schedule in the day-care center included psychological treatment, nutritional and psychiatric supervision, parental counseling, and various group therapies carried out in the hospital setting three times a week. To prevent rehospitalization, Y.s' parents were encouraged to increase their involvement with her eating at home.

At the break of the COVID-19 pandemic, Y. was on the waiting list for recurrent inpatient treatment. At that time, all the daycare services had to be changed to telemedicine, including her once-weekly psychiatric, nutritional, psychological and parental counseling. Group therapy with other patients from the daycare treatment was also performed via telemedicine once a week. Y. was weighed once weekly in the hospital, this being the only

face-to-face intervention. She continued with daily food monitoring with the assistance of her parents.

As Y, was at home with her family, this allowed for the augmentation of family interventions. The aim of this treatment was to use the opportunity of the COVID-19 confinement to improve the communication between Y. and her parents, and to empower her parents in assisting her with her eating. The parents, themselves, were preoccupied during the lockdown with their own physical training and dieting. In the sessions with the psychologist, Y. shared that her parents could not assist with her eating because of their own physical activities and dieting, thus refusing to cooperate with their supervision. The parents' behaviors were brought to the foreground and discussed in the telemedicine parental counseling, leading to their willingness to change them with the assistance of the treatment team. This, in turn, led Y. to agree to eat under her parents' supervision, gradually gaining weight again, eventually reaching her target weight. After the COVID-19 contingencies of the first quarantine were removed, Y. was able to resume daycare treatment, nor requiring re-hospitalization anymore. Currently she is in ambulatory treatment, with all sessions carried out face-to-face.

# Inpatient treatment: case 4

S. is 17 years old, a 12th grade female student, the second of two children, living with her parents. There is a family history of anxiety disorders. She is diagnosed with social anxiety disorder from an early age.

S. developed AN-restricting type with comorbid major depressive disorder in the past three years. Combined with her pre-existing social anxiety disorder, she stopped attending school and later did not leave her room for prolonged periods. S. was hospitalized in our center after a year of unsuccessful ambulatory treatment for her ED. When the COVID-19 outbreak started, she was already hospitalized for four months. At that time, S. reached her target weight range and her depression was in remission. Although she was planned to be released gradually from inpatient treatment to daycare, the COVID-19 outbreak required her immediate discharge to virtual telemedicine home-based treatment. This included once-weekly individual psychotherapy and once-weekly parental, nutritional and psychiatric counseling, alongside the provision of school services via telemedicine three days a week. S. was weighed every two weeks in the hospital by the department's nurse. This new treatment program continued for three months.

Despite this intensive multi-professional treatment, the change of the treatment setting was followed by a continuous deterioration in S.'s condition. This included the returning of restriction of food intake and of previous maladaptive eating-related behaviors and social

withdrawal. S. refused to take her medications (fluoxetine 20 mg/day and risperdone 1 mg/day) and shared with the staff that she had difficulties coping without the support of the inpatient setting. As a result, the team decided to augment the parental telemedicine counseling sessions to twice weekly and long-distance schooling to five days weekly.

These changes were possible because of the motivation of the treatment staff to be flexible and treat after the regular working hours. The flexible use of telemedicine with different team providers and different family members, at different times, enabled the carrying out of this plan. Indeed, the family assisted S. in reducing her avoidant strategies, and regain her studying - first in the department's long-distance classes and thereafter, when the mandatory lockdown was released, to gradually return to her previous school, alongside an improvement in her eating behaviors and weight. Overall, the use of intensive home-based multidisciplinary telemedicine during the pandemic quarantine allowed for the integration of S. back into our daycare program and her school when the COVID-19 contingencies were released. Currently she continues with ambulatory treatment, performed totally with face-to-face interventions.

# Discussion

# Overall data

The COVID-19 outbreak in Israel has developed rapidly and unexpectedly from March 2020. The analysis of the treatment data in our ambulatory service during the first 10 months of 2020 shows greater use of mental health service during the COVID-19 outbreak and enforced quarantine in adolescents with EDs in comparison to regular times. The increase in the number of sessions has been likely accounted for, at least in part, by using multi-professional telemedicine meetings, comprising of 37% of all sessions during the first 10 months of 2020, vs. no use in the respective period in 2015-2019 (see Table 1). It is of importance to note, with respect to the considerable use of telemedicine in our study, that our team, like any other team in the hospital, has been instructed by specific professionals about the use of telemedicine, both technically and regarding its advantages and risks. This may have accounted for the overall positive reactions of most team members from all professions to using telemedicine in the treatment of adolescents with EDs, although being mostly not familiar with its use previously.

Our findings are in line with a study performed in the United States and Netherlands [10], showing that 42–45% of adult patients with EDs have transitioned to online/telehealth care during the COVID-19 pandemic. They are also similar to a recent study in adolescents with EDs in Spain [19], reporting more visits, but a

lower number of patients, during the COVID-19 confinement compared with the same period in the previous year. In that study, the treatment team held 1329 (73.10%) telehealth consultations and 489 (26.9%) face-to-face outpatient visits with 365 patients undergoing ambulatory or daycare treatment. Nonetheless, another study of ambulatory adolescents with EDs in Singapore showed a decrease of 50% in treatment sessions during the COVID-19 period [18]. Our findings also stand in contrast to a survey of adult patients with BN in Germany, reporting that face-to-face psychotherapy decreased by 56% during the COVID-19 pandemic, and videoconferencing therapy was used only by 22% of the patients [13].

Several explanations can be suggested for these discrepancies. In contrast to other studies, telemedicine was not used in our service before the COVID-19 breakout. Second, in the first lockdown in Israel, everyone stayed at home, both parents and children. This could have increased tension and distress, and in turn, also ED-related symptoms, altogether requiring more intensive treatment [4–7, 10–15]. Thus, the COVI9–19 pandemic with its strict mandatory confinement might have created an environment that developed tougher-to treat EDs. This was highlighted in our study by the greater number of sessions administered at that time with only a slight decrease in the number of patients using them (see Table 1). At the same time, the provision of telemedicine, with all family members confined to their home, enabled this greater use of treatment. Overall, these conditions likely increased the cost of care, despite the cost of telemedicine being lower than that of faceto-face interventions.

Our data is similar, in this respect, to findings in Spain, another country requiring strict confinement at the start of the COVID-19 [19], leading to greater usage of long-distance treatment. Other telemedicine- associated factors contributing to the increase in the number of treatment sessions in our study can be related to the high flexibility in the scheduled sessions, and to less cancellation of meetings from patients and treatment providers.

#### Case reports

The condition of our four female adolescents with AN has been compromised at the start of the COVID-19 quarantine. Changes in physical activity, sleep patterns, screen time and eating have been reported in healthy children and adolescents during periods without school, when spending most of the time at home [44]. These effects may be exacerbated when children are confined to their homes, causing them to have little contact with their peers and depriving them of outdoor activities [19]. However, the specific changes in the routine in

adolescents with EDs, as described in our cases, seem to differ, showing an increase in physical activity rather than the decrease shown in healthy youngsters, and restricting rather than an increase in the amount of food eaten [19]. These changes in lifestyle may likely increase the risk of ED-related symptomatic deterioration [19], alongside the influence of acute depressive, anxiety, and trauma- related symptoms resulting from the COVID-19 outbreak, with its consequent quarantine and social isolation [4–6, 18].

In line with these findings, one study of adolescents with EDs treated in an ambulatory service in Singapore reported of an increase in health-related anxiety associated with the COVID-19 outbreak [18]. By contrast, our four adolescents hardly related to the pandemic by itself; they were mostly concerned with what was important for them personally [45]. Not surprisingly, this was associated with the influence of the quarantine-related change in their life conditions on their weight and eating/weight-related behaviors, as well as with the greater unwanted involvement of their families with their eating [see cases (1,), (2,), and 3).

Despite their seeming indifference to the direct healthrelated concerns of the pandemic, it is plausible to assume that because of their constricted, inflexible, rigid, and self-centered cognitive style [46, 47], our adolescents with AN may have been affected to a greater extent than healthy youngsters by the COVID-19 outbreak and its psychosocial consequences. Moreover, in adolescents with AN, the inability to escape the parents' watching eyes at every meal during the confinement and to go outside and exercise, as described in the second ambulatory case, may increase their overall distress and distrust of others [47], adversely impacting their cooperation with treatment. Resulting fears of loss of personal control may trigger an increase in weight control behaviors in these adolescents to compensate for this loss [48]. This regression has been shown to occur, to some extent, in all four cases, although it could be improved in three of the four [Case (1, 3, 4)], following the intervention of the team with the eating-related involvement of their families.

Alternatively, youngsters with more severe AN and greater overall psychopathology [47], as described in Case (4), may have been isolated from their peers to some extent already before the COVID-19 pandemic. Along these lines, the absence during the COVID-19 quarantine in these youngsters of intense weight-related comparison driven by social contact [49], and the reduction in social stressors that hitherto increased their social-related anxieties [9, 49], might have actually reduced their overall distress. This has likely happened with Case (3) during her prolonged inpatient treatment, but not when staying at home during the COVID-19

quarantine, when losing the safe environment of the department.

#### Telemedicine

Treatment of adolescents with psychiatric disorders in general, and EDs in particular, during the severe, unexpected changes of the COVID-19 and its associated quarantine, is highly challenging, requiring multidisciplinary collaboration, often with hitherto unfamiliar solutions. This is of importance, because in a recent study of adolescents with diverse psychiatric disturbances, most reported not only worsening of their mental condition during the COVID-19 pandemic, but a quarter stopped using mental health support [50]. This finding is very concerning, specifically for adolescents with EDs, who are often in need for multidisciplinary psychological, familial, nutritional and medical interventions [7].

In contrast to regular times, telemedicine in the COVID-19 period was not a matter of preferences by the patient/family or treatment providers. All treatment requirements, including weight measurements, had to be carried out online; all family members were at home, interfering with the patient's privacy, the latter being especially important for adolescents with AN. In addition, families suffered from health- and economic-related preoccupations not present before [19, 30].

Several specific issues have to be considered regarding the telemedicine treatment of adolescents with EDs during the COVID-19 period. Thus, Rodgers et al. [51] have found that video conferencing in these patients may exacerbate body image concerns by increasing their preoccupation with and focusing on their self-appearance. In contrast, there is the issue of the "disembodied environment", when only the patient's and therapist's faces, but not their bodies, appear on the screen, or when the patient prefers to converse in the session with the camera being closed. What is potentially missing in these cases is body-to-body communication, or the reading of body language [52].

In the present study we had the opportunity to learn from our own experience about the pros and cons of long-distance interventions in the treatment of adolescents with EDs. From the positive aspect, the telemedicine interventions were found to flexible, relatively easily and quickly arranged in times of crisis, and enabled the provision of treatment in hours that were more suitable for both treatment consumers and providers. Children did not have to leave school when the conditions of the confinement subsided, parents did not have to lose working hours, and other family members could relatively easily join treatment whenever required [see cases (1) and (2)]. Team meetings of different team members were relatively easily planned and carried out, including in the therapists' homes. Moreover, the social distancing

Yaffa et al. Journal of Eating Disorders (2021) 9:17 Page 9 of 11

and "disembodied environment" generated with telemedicine was likely adequate for at least in three of the four cases [cases (1), (3) and (4)]. This is not surprising, bearing in mind the high rate of comorbid social anxieties in patients with EDs [53], and their concern with face-to-face emotional interactions, often considered by them to be overwhelming [54].

In these three cases, the families were quite well organized and flexible to work with telemedicine with the treatment providers according to the interventions recommended. In Case (1), this was despite the divorce of the parents, the lack of involvement of the father and both partners of the parents, and the unexpected return of the older siblings to the mother's house because of the COVID-19 outbreak. The flexibility of the siblings in being able to change, following the team's recommendation, their role from supervising the girl's eating to empower the mother in her supervision, was of particular merit. In Case (3), the parents could change, with the assistance of the team, from being preoccupied with their own COVID-19 related problems to concentrate on the meal plan of their daughter. The augmentation of the parental telemedicine counseling sessions to twice weekly in Case (4) had a big influence on her cooperation with the parental supervised meal plan. These preliminary clinical findings suggest that adolescents with AN living in adaptive families can be managed with telemedicine in pandemic quarantine conditions. Nonetheless, it is of note that even in these favorable conditions, self-weighing, weighing in front of a family member, and meal supervision was difficult to implement.

Long-distance telemedicine consultation and psychotherapy was not adequate for Case (2), likely because of the lack of involvement of both parents with the treatment. The mother has been mostly confined to a helpless position, the father has not been involved, and telemedicine has not been enough to empower other family members (siblings). Our findings in this case, thus support the suggestion of Guo et al. [55], that families experiencing more problems may need more online intensive treatment or face-to face meetings with their treatment providers to assist their children. Interestingly, in this study [55], caregivers of patients with EDs have been found to show higher levels of depression and anxiety than non-ED caregivers, likely interfering with their potential to assist their patients. Moreover, a brief online education program has shown no effect on decreasing the depression and anxiety of the caregivers.

To summarize, the four cases described here highlight the important role of the family in improving, or alternatively maintaining, the adolescent's illness [56], and in cooperating with the treatment of their daughter. This may become even more crucial in times of crises such as the COVID-19 quarantine, when the whole family is at home for prolonged periods, and has to be treated with long-distance interventions. Future studies of adolescents with EDs during pandemic times such as the COVID-19 outbreak, comparing between telemedicine, face-to-face and combined interventions with the use of systematic interventions and standardized assessment tools, should take into consideration the functioning of the family in the choice of the suitable mode of treatment for the specific adolescent and family.

#### **Conclusion**

The condition of our four adolescents with AN was likely compromised at the start of the COVID-19 outbreak. Patients and families had to change their routines within a brief period and adjust to the new circumstances characterized by higher presence and greater involvement of family members with the adolescent's eating, decrease in the options for physical activities, and lower access to treatment. The use of telemedicine treatment, which in our facilities was limited before the pandemic, offered new possibilities, including flexible multidisciplinary treatment provision with different treatment providers, the inclusion of different family members in different treatment sessions, and relatively easily planned team consultations.

It is of note, that similar telemedicine interventions were introduced in our facility regardless of the level of care, i.e., ambulatory, daycare, or inpatient interventions. This enabled ambulatory patients to receive more multidisciplinary treatment sessions during the COVID-19 outbreak than before, and daycare and inpatients to receive personally adjusted flexible programs in comparison to the more standardized interventions provided in regular times. The favorable effects of telemedicine were exemplified by the large number of patients who chose to continue with it even when face-to-face services were resumed.

Finally, we hope that the reported information of our case series can assist clinicians in their management of their ED patients and contribute to the limited literature on the treatment of adolescents with EDs during the still ongoing COVID-19 pandemic.

#### Abbreviations

EDs: Eating disorders; AN: Anorexia nervosa; BN: Bulimia nervosa; BMI: body mass index

#### Acknowledgements

Not applicable.

#### Authors' contributions

Drs. Serur, Enoch-Levy and Stein wrote the introduction, case reports and discussion. Drs. Pesach and Gothelf initiated the idea of the article and read and corrected all versions. Ms. Yoffe-Milstein contributed to the discussion about the use of telemedicine. The authors read and approved the final manuscript.

#### **Funding**

There was no funding.

# Availability of data and materials

Not applicable.

# Ethics approval and consent to participate

Due to the nature of this case report, ethics approval was not required by the institution.

#### Consent for publication

Demographic and clinical details of the participants were changed to prevent identification. Verbal consent was obtained for publication from parents and patients.

#### Competing interests

The authors declare that they have no competing interests.

#### **Author details**

<sup>1</sup>Pediatric Psychosomatic Department, Sheba Medical Center, 5265601 Tel Hashomer, Israel. <sup>2</sup>Child and Adolescent Psychiatric Division, Sheba Medical Center, 5265601 Tel Hashomer, Israel. <sup>3</sup>Edmond and Lily Safra Children's Hospital, Sheba Medical Center, 5265601 Tel Hashomer, Israel. <sup>4</sup>Sackler Faculty of Medicine, Tel Aviv University, 69978 Tel Aviv, Israel. <sup>5</sup>Sagol School of Neuroscience, Tel Aviv University, 69978 Tel Aviv, Israel.

# Received: 28 August 2020 Accepted: 29 January 2021 Published online: 10 February 2021

#### References

- Smith K, Ostinelli E, Macdonald O, Cipriani A. COVID-19 and telepsychiatry: an evidence-based guidance for clinicians. JMIR Ment Health. 2020;7(8): e21108.
- O'Hara VM, Johnston SV, Browne NT. The paediatric weight management office visit via telemedicine: pre-to post-COVID-19 pandemic. Pediatric Obesity. 2020;15(8):e12694.
- Aldao A, Nolen-Hoeksema S, Schweizer S. Emotion-regulation strategies across psychopathology: a meta-analytic review. Clin Psychology Rev. 2010; 30(2):217–37.
- Fiorillo A, Gorwood P. The consequences of the COVID-19 pandemic on mental health and implications for clinical practice. Eur Psychiatry. 2020; 63(1):e32
- Hawryluck L, Gold WL, Robinson S, Pogorski S, Galea S, Styra R. SARS control and psychological effects of quarantine, Toronto. Canada Emerg Infect Dis. 2004;10(7):1206–12.
- Guessoum SB, Lachal J, Radjack R, Carretier E, Minassian S, Benoit L, et al. Adolescent psychiatric disorders during the covid-19 pandemic and lockdown. Psychiatry Res. 2020;291:113264.
- 7. Mitchell JE, Peterson CB. Anorexia nervosa. N Engl J Med. 2020;382:1343–51.
- Sabel AL, Gaudiani JL, Statland B, Mehler PS. Hematological abnormalities in severe anorexia nervosa. Ann Hematol. 2013;92(5):605–13.
- Walsh O, McNicholas F. Assessment and management of anorexia nervosa during COVID-19. Ir J Psychol Med. 2020;37(3):187–91.
- Termorshuizen JD, Watson HJ, Thornton LM, Borg S, Flatt RE, MacDermod CM, et al. Early Impact of COVID-19 on individuals with eating disorders: A survey of~ 1000 individuals in the United States and the Netherlands. Int J Eat Disord. 2020;53(11):1780–90.
- Matheson BE, Bohon C, Lock J. Family-based treatment via videoconference: clinical recommendations for treatment providers during COVID-19 and beyond. Int J Eat Disord. 2020;53(7):1142–54.
- Branley-Bell D, Talbot CV. Exploring the impact of the COVID-19 pandemic and UK lockdown on individuals with experience of eating disorders. J Eat Disord. 2020;8:44.
- Schlegl S, Meule A, Favreau M, Voderholzer U. Bulimia nervosa in times of the COVID-19 pandemic—results from an online survey of former inpatients. Eur Eat Disord Rev. 2020;28(6):847–54.
- Clark Bryan D, Macdonald P, Ambwani S, Cardi V, Rowlands K, Willmott D, Treasure J. Exploring the ways in which COVID-19 and lockdown has affected the lives of adult patients with anorexia nervosa and their carers. Eur Eat Disord Rev. 2020;28(6):826–35.

- Fernández-Aranda F, Casas M, Claes L, Bryan DC, Favaro A, Granero R, et al. COVID-19 and implications for eating disorders. Eur Eat Disord Rev. 2020; 28(3):239–45.
- Touyz S, Lacey H, Hay P. Eating disorders in the time of COVID-19. J Eat Disord. 2020;8:19.
- Cooper M, Reilly EE, Siegel JA, Coniglio K, Sadeh-Sharvit S, Pisetsky EM, et al. Eating disorders during the COVID-19 pandemic and quarantine: an overview of risks and recommendations for treatment and early intervention. Eat Disord. 2020;9:1–23.
- Courtney D, Kee CN, Yin OJ, Amerie B, Kumudhini R, Shan EC. Caring for children and adolescents with eating disorders in the current COVID-19 pandemic: a Singapore perspective. J Adolesc Health. 2020;67(1):131–4 https://www.ncbi.nlm.nih.gov/pmc/articles/PMC7136888/.
- Graell M, Morón-Nozaleda MG, Camarneiro R, Villaseñor Á, Yáñez S, Muñoz R, Martínez-Núñez B, Miguélez-Fernández C, Muñoz M, Faya M. Children and adolescents with eating disorders during COVID-19 confinement: difficulties and future challenges. Eur Eat Disord Rev. 2020;28(6):864–70.
- Bokolo AJ. Exploring the adoption of telemedicine and virtual software for care of outpatients during and after COVID-19 pandemic. Ir J Med Sci. 2020; 8:1–0
- Barak A, Hen L, Boniel-Nissim M, Shapira NA. A comprehensive review and a meta-analysis of the effectiveness of internet-based psychotherapeutic interventions. J Technol Hum Serv. 2008;26(2–4):109–60.
- 22. Andersson G. Internet interventions: past, present and future. Internet Interv. 2018;12:181–8.
- Carlbring P, Andersson G, Cuijpers P, Riper H, Hedman-Lagerlöf E. Internetbased vs. face-to-face cognitive behavior therapy for psychiatric and somatic disorders: an updated systematic review and meta-analysis. Cogn Behav Ther. 2018;47(1):1–8.
- Castro A, Gili M, Ricci-Cabello I, Roca M, Gilbody S, Perez-Ara MÁ, Seguí A, McMillan D. Effectiveness and adherence of telephone-administered psychotherapy for depression: a systematic review and meta-analysis. J Affect Disord. 2020;260:514–26.
- Liu L, Thorp SR, Moreno L, Wells SY, Glassman LH, Busch AC, Zamora T, Rodgers CS, Allard CB, Morland LA, Agha Z. Videoconferencing psychotherapy for veterans with PTSD: results from a randomized controlled non-inferiority trial. J Telemed Telecare. 2020;26(9):507–19.
- Berryhill MB, Culmer N, Williams N, Halli-Tierney A, Betancourt A, Roberts H, King M. Videoconferencing psychotherapy and depression: a systematic review. Telemedicine and e-Health. 2019;25(6):435–46.
- 27. Berryhill MB, Halli-Tierney A, Culmer N, Williams N, Betancourt A, King M, Ruggles H. Videoconferencing psychological therapy and anxiety: a systematic review. Fam Pract. 2019;36(1):53–63.
- Backhaus A, Agha Z, Maglione ML, Repp A, Ross B, Zuest D, Rice-Thorp NM, Lohr J, Thorp SR. Videoconferencing psychotherapy: a systematic review. Psychol Serv. 2012;9(2):111.
- Anastasiadou D, Folkvord F, Brugnera A, Cañas Vinader L, Serrano Troncoso E, Carretero Jardí C, Linares Bertolin R, Muñoz Rodríguez R, Martínez Nuñez B, Graell Berna M, Torralbas-Ortega J. An mHealth intervention for the treatment of patients with an eating disorder: a multicenter randomized controlled trial. Int J Eat Disord. 2020;53(7):1120–31.
- Brooks SK, Webster RK, Smith LE, Woodland L, Wessely S, Greenberg N, Rubin GJ. The psychological impact of quarantine and how to reduce it: rapid review of the evidence. Lancet. 2020;395(10227):912–20.
- 31. Liu S, Yang L, Zhang C, Xiang YT, Liu Z, Hu S, Zhang B. Online mental health services in China during the COVID-19 outbreak. Lancet Psychiatry. 2020;7(4):e17–8.
- 32. Weinberg H. Online group psychotherapy: Challenges and possibilities during COVID-19—A practice review. Group Dynamics Theory Res Pract. 2020;24(3):201.
- 33. Zhou X, Snoswell CL, Harding LE, Bambling M, Edirippulige S, Bai X, Smith AC. The role of telehealth in reducing the mental health burden from COVID-19. Telemedicine and e-Health. 2020;26(4):377–9.
- Dimitropoulos G, Landers A, Freeman V, Novick J, Schmidt U, Olmsted M. A feasibility study comparing a web-based intervention to a workshop intervention for caregivers of adults with eating disorders. Eur Eat Disord Rev. 2019;27(6):641–54.
- Pierce BS, Perrin PB, McDonald SD. Path analytic modeling of psychologists' openness to performing clinical work with telepsychology: a national study. J Clin Psychol. 2020;76(6):1135–50.

- 36. American Psychiatric Association. Practice guideline for the treatment of patients with eating disorders. Am J Psychiatry. 2006;163:4–54.
- Anzai N, Lindsey-Dudley K, Bidwell RJ. Inpatient and partial hospital treatment for adolescent eating disorders. Child Adolesc Psychiatr Clin N Am. 2002;11(2):279.
- Lock J, Le Grange D. Family-based treatment: where are we and where should we be going to improve recovery in child and adolescent eating disorders. Int J Eat Disord. 2019;52(4):481–7.
- Cook-Darzens S, Doyen C, Mouren MC. Family therapy in the treatment of adolescent anorexia nervosa: current research evidence and its therapeutic implications. Eat Weight Disord. 2008;13(4):157–70.
- Fairburn CG, Marcus MD, Wilson GT. Cognitive-behavioral therapy for binge eating and bulimia nervosa: a comprehensive treatment manual. In: Fairburn CG, Wilson TG editors. Binge eating: Nature, assessment and treatment. New York: Guilford Press; 1993. pp 361–404.
- Schmidt U, Wade TD, Treasure J. The Maudsley model of anorexia nervosa treatment for adults (MANTRA): development, key features, and preliminary evidence. J Cogn Psychother. 2014;28(1):48–71.
- 42. Sansone RA, Levitt JL, Sansone LA. A primer on psychotherapy treatment of anorexia nervosa in adolescents. Psychiatry (Edgmont). 2005;2(2):40.
- Abbate-Daga G, Marzola E, De-Bacco C, Buzzichelli S, Brustolin A, Campisi S, Amianto F, Migliaretti G, Fassino S. Day hospital treatment for anorexia nervosa: a 12-month follow-up study. Eur Eat Disord Rev. 2015;23:390–8.
- Wang G, Zhang Y, Zhao J, Zhang J, Jiang F. Mitigate the effects of home confinement on children during the COVID-19 outbreak. Lancet. 2020; 395(10228):945–7.
- 45. Fox A, Harrop C, Trower P, Leung N. A consideration of developmental egocentrism in anorexia nervosa. Eat Behav. 2009;10(1):10–5.
- Friederich HC, Herzog W. Cognitive-behavioral flexibility in anorexia nervosa. Curr Top BehavNeurosci. 2011;6:111–23.
- Amianto F, Abbate-Daga G, Morando S, Sobrero C, Fassino S. Personality development characteristics of women with anorexia nervosa, their healthy siblings and healthy controls: what prevents and what relates to psychopathology? Psychiatry Res. 2011;187(3):401–8.
- Tiggemann M, Raven M. Dimensions of control in bulimia and anorexia nervosa: internal control, desire for control, or fear of losing self-control? Eat Disord. 1998;6(1):65–71.
- Haynos AF, Watts AW, Loth KA, Pearson CM, Neumark-Stzainer D. Factors predicting an escalation of restrictive eating during adolescence. J Adolesc Health. 2016;59(4):391–6.
- Nathiya D, Singh P, Suman S, Raj P, Tomar BS. Mental health problems and impact on youth minds during the COVID-19 outbreak: cross-sectional (RED-COVID) survey. Soc Health Behav. 2020;3(3):83–8.
- Rodgers RF, Lombardo C, Cerolini S, Franko DL, Omori M, Fuller-Tyszkiewicz M, Linardon J, Courtet P, Guillaume S. The impact of the COVID-19 pandemic on eating disorder risk and symptoms. Int J Eat Disord. 2020; 53(7):1166–70.
- 52. Ogden P, Goldstein B. Sensorymotor psychotherapy from a distance. In: Weinberg H, Rolnick A, editors. Engaging the body, creating presence and building relationship in videoconferencing. Theory and practice of online therapy: internet-delivered interventions for individuals, families, groups, and organizations. New York: Routledge. 2020;5:47–63.
- Kaye WH, Bulik CM, Thornton L, Barbarich N, Masters K. Comorbidity of anxiety disorders with anorexia and bulimia nervosa. Am J Psychiatry. 2004; 161(12):2215–21.
- Heilbrun AB Jr, Worobow AL. Attention and disordered eating behavior: II.
  Disattention to turbulent inner sensations as a risk factor in the development of anorexia nervosa. Psychol Rep. 1990;66(2):467–78.
- Guo L, Wu M, Zhu Z, Zhang L, Peng S, Li W, Chen H, Fernández-Aranda F, Chen J. Effectiveness and influencing factors of online education for caregivers of patients with eating disorders during COVID-19 pandemic in China. Eur Eat Disord Rev. 2020;28(6):816–25.
- Rienecke RD, Lebow J, Lock J, Le Grange D. Family profiles of expressed emotion in adolescent patients with anorexia nervosa and their parents. J Clin Child Adolesc Psychol. 2017;46(3):428–36.

# **Publisher's Note**

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

# Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

# At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

